<mark>,</mark>



# **Geophysical Research Letters**<sup>\*</sup>

# **RESEARCH LETTER**

10.1029/2023GL104347

#### **Key Points:**

- A deep-learning model outperforms dynamic models, filling a subseasonal Antarctic sea ice prediction gap by bypassing physical mechanisms
- Unlike a Markov model that predicts time series of fixed spatial patterns, our model can extract ice synchronized spatiotemporal evolutions
- The deep-learning model captures climate signals in sea ice, delivering the best prediction in fall season and the Weddell Sea region

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### Correspondence to:

X. Li and X. Yuan, lixf@qdio.ac.cn; xyuan@ldeo.columbia.edu

#### Citation:

Wang, Y., Yuan, X., Ren, Y., Bushuk, M., Shu, Q., Li, C., & Li, X. (2023). Subseasonal prediction of regional Antarctic sea ice by a deep learning model. *Geophysical Research Letters*, 50, e2023GL104347. https://doi. org/10.1029/2023GL104347

Received 2 MAY 2023 Accepted 18 AUG 2023

© 2023. The Authors. This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

# Subseasonal Prediction of Regional Antarctic Sea Ice by a Deep Learning Model

Yunhe Wang<sup>1</sup>, Xiaojun Yuan<sup>2</sup>, Yibin Ren<sup>1</sup>, Mitchell Bushuk<sup>3</sup>, Qi Shu<sup>4</sup>, Cuihua Li<sup>2</sup>, and Xiaofeng Li<sup>1</sup>

<sup>1</sup>CAS Key Laboratory of Ocean Circulation and Waves, Institute of Oceanology, Chinese Academy of Sciences, Qingdao, China, <sup>2</sup>Lamont-Doherty Earth Observatory of Columbia University, Palisades, New York, USA, <sup>3</sup>National Oceanic and Atmospheric Administration/Geophysical Fluid Dynamics Laboratory, Princeton, NJ, USA, <sup>4</sup>First Institute of Oceanography, Ministry of Natural Resources, Qingdao, China

**Abstract** Antarctic sea ice concentration (SIC) prediction at seasonal scale has been documented, but a gap remains at subseasonal scale (1–8 weeks) due to limited understanding of ice-related physical mechanisms. To overcome this limitation, we developed a deep learning model named Sea Ice Prediction Network (SIPNet) that can predict SIC without the need to account for complex physical processes. Compared to mainstream dynamical models like European Centre for Medium-Range Weather Forecasts, National Centers for Environmental Prediction, and Seamless System for Prediction and Earth System Research developed at Geophysical Fluid Dynamics Laboratory, as well as a relatively advanced statistical model like the linear Markov model, SIPNet outperforms them all, effectively filling the gap in subseasonal Antarctic SIC prediction capability. SIPNet results indicate that autumn SIC variability contributes the most to sea ice predictability, whereas spring contributes the least. In addition, the Weddell Sea displays the highest sea ice predictability, while predictability is low in the West Pacific. SIPNet can also capture the signal of ENSO and SAM on sea ice.

**Plain Language Summary** Antarctic sea ice has changed significantly since 2016, leading to a higher demand for sea ice forecasts. However, forecasting Antarctic sea ice has not received enough attention. Limited observations and lack of understanding of ice-related physical mechanisms result in significant errors in sea ice predictions in dynamical models, particularly at the subseasonal timescale. We utilized a deep-learning model to predict Antarctic sea ice at this timescale to fill this gap. Results showed that the deep-learning model performed skillfully 1–8 weeks in advance, with the Weddell Sea being the best-predicted region and the West Pacific being the worst. Furthermore, our study found that the model significantly outperformed mainstream dynamic models and a conventional statistical model. These findings build a foundation for developing more advanced prediction models at high resolutions for operational applications.

#### 1. Introduction

In contrast to the rapid sea ice losses reported in the Arctic, the Antarctic sea ice extent (SIE) showed a moderate upward trend from 1979 to 2015 amidst anthropogenic warming (Comiso et al., 2017; Parkinson & Cavalieri, 2012), followed by a significant recent decline (Fogt et al., 2022; Parkinson, 2019; Zhang et al., 2022). Antarctic sea ice exhibits marked regional variability with strong zonal asymmetry (Li et al., 2021; Yuan et al., 2018) and a strong seasonal cycle (Handcock & Raphael, 2020). It appears to be more volatile in the recent decade, including record highs and lows, which leads to increasing demand for Antarctic sea ice forecasts.

While much attention has been given to predictions of Arctic sea ice, predictions of Antarctic sea ice have received relatively fewer assessments and remain a substantial challenge, especially at the subseasonal timescale. For instance, the most recent results from the Sea Ice Prediction Network South (SIPN South) show that Antarctic sea ice forecasts are not yet accurate enough to guide field planning or maritime route forecasting and are still at a very early stage (Massonnet et al., 2022; Zampieri et al., 2019). In addition, dynamic models have challenges in parameterizing various processes such as the sea ice thickness (SIT) distribution, wave-ice interaction, rheology, land-fast ice, and floe size distribution (Leppäranta et al., 2020; Zampieri et al., 2018). Lack of in-depth knowledge of these processes and errors in modeling ocean and atmosphere interactions led to a decline of Antarctic SIE during the satellite era in model simulations, contrary to the statistically significant increase in observations

19448007, 2023, 17, Downloa

1029/2023GL104347 by Co

Wile

Online Library on [31/08/2023].

(Roach et al., 2020). Moreover, the inter-model spread in SIPN South is much larger than the observational uncertainty (Massonnet et al., 2022).

Although earlier investigations have made some progress in examining the seasonal sea ice prediction in the Antarctic, revealing the significant influence of upper-ocean heat content on winter sea-ice forecasts (Bushuk et al., 2021; Marchi et al., 2019) and the crucial role of sea-ice thickness in predicting summer sea ice (Bushuk et al., 2021; Morioka et al., 2021), research on the subseasonal scale (1–8 weeks) is limited and challenging. Zampieri et al. (2019) found that the subseasonal forecasting skill of the Antarctic sea ice edge is 30% lower than that of the Arctic.

Given the paucity of observations and challenges in simulating sea ice physics in the Antarctic, is it feasible to pursue an alternative approach by utilizing deep-learning (DL) methodology for sea ice forecasting at the subseasonal scale? By extracting sea ice spatiotemporal features at multiple scales, DL has an immense potential to capture signals of sea ice predictability and avoid errors caused by incomplete parameterization in the complicated ocean-atmosphere-ice system (Andersson et al., 2021; Chi & Kim, 2017; Kim et al., 2020; Liu et al., 2021). In this study, we develop a DL model called sea ice prediction network (SIPNet) to predict subseasonal Antarctic sea ice concentration (SIC) using only SIC as input. The aim is to examine whether the DL model can significantly outperform dynamical and statistical models and fill the gap in Antarctic sea ice forecasts at the subseasonal timescale.

#### 2. Data and Methods

#### 2.1. Data

We used daily bootstrap SICs from 1979 to 2021 obtained from the National Snow and Ice Data Center (NSIDC) (Comiso, 2017) to train our DL model and evaluate its performance. The data set was derived from passive microwave radiometers with a spatial resolution of 25 km. To examine possible data dependencies, we conducted a comparative analysis utilizing daily SICs from the Ocean and Sea Ice Satellite Application Facility (OSI SAF) version 3 (Sørensen et al., 2022), with identifiers OSI-430-a (1979–2020) and OSI-450-a (2021). The data set was derived from passive microwave radiometers in conjunction with European Centre for Medium-Range Weather Forecasts (ECMWF) reanalysis version 5 (ERA5). SIC predictions from ECMWF and the National Centers for Environmental Prediction (NCEP) database of the subseasonal to seasonal (S2S) Prediction project (Vitart et al., 2017) and the Seamless System for Prediction and Earth System Research developed at Geophysical Fluid Dynamics Laboratory (GFDL-SPEAR; Delworth et al., 2020) were used for the comparative analysis. Additionally, Southern Annular Mode (SAM; Marshall, 2003) and Niño 3.4 indices (Rayner et al., 2003) were utilized to assess the SIPNet's ability to capture climate-variability signals in sea ice forecasting.

#### 2.2. Methods

Unlike dynamical models that rely on coupling processes among sea ice, atmosphere, and ocean, the SIPNet only uses SIC data for training and forecasting since sea ice variability is an integrated result of atmosphere-ocean-sea ice interactions. SIPNet model composes a multi-scale nested encoder-decoder structure and mainly includes four modules: input, encoder, decoder, and output (Figure 1a). We use the historical SIC of the last 8 weeks to predict the following 8 weeks. The input is the previous 8-week SIC sequence with dimensions  $304 \times 320 \times 8$ , which has been cropped in the spatial dimensions. The reason and method for data cropping can be found in Text S1 in Supporting Information S1. The encoder reduces the spatial dimensions of the input SIC sequences by max-pooling and increases the number of feature maps by convolution layers (Ren & Li, 2023; Ren et al., 2022). This process captures spatiotemporal connections from different down-scaled feature maps. The decoder consists of a series of upsampling and convolutional layers that increase the spatial dimensions of feature maps and reduce the number of sequences. At each stage, concatenations are employed to connect the feature map acquired by the encoder and multi-scale nested decoder at the same level, facilitating the integration of spatial dependencies across multiple scales. The feature maps output of the decoder is directed to the output module, which comprises a single convolution neural network (CNN) layer with eight  $1 \times 1$  convolutional kernels. A sigmoid activation layer is subsequently applied to the convolutional outputs, predicting the SIC in each grid cell. The loss is computed by comparing the predicted SIC with the observed SIC. By minimizing the losses step-by-step, the model is trained to a well-performed state. The model uses zero padding to ensure that the output maps have the

# **Geophysical Research Letters**



**Figure 1.** Structure of SIPNet model and Antarctic regions used in this study. (a) Model structure including input, encoder, decoder, and output; input set with weekly SICs; encoder module comprises the ResNet<sub>TSAM</sub> and max-pooling layer; decoder is composed of ResNet<sub>TSAM</sub>, upsampling, and concatenation layer; output set with predicted weekly SICs of corresponding lead weeks. (b) Structure of the ResNet<sub>TSAM</sub> block. (c) Antarctic regions; the blue line represents the maximum extent of sea ice from 1979 to 2021, which is used as the study area. "A and B" represents the Amundsen and Bellingshausen Seas.

same spatial dimensions as the input SIC. More detailed information regarding the model is provided in Text S1 in Supporting Information S1.

The SIC data set spanning 43 years was partitioned into three sets: training (1979–2010), model validation (2011–2016), and assessment (2017–2021). Compared to daily data, weekly average SIC has a higher signal-to-noise ratio and longer persistence. To address the inadequate weekly time series for model training, we apply data preprocessing before input (Figure 1a), which employs a strategy of expanding the weekly time series from daily data. See the Input section of Text S1 in Supporting Information S1 for details. These created weekly SIC time series are only employed for model training and validation. Model assessment is conducted using standard weekly data.

In addition, the forecasting results were evaluated seasonally, with the seasons defined as winter (June to August), spring (September to November), summer (December to February), and autumn (March to May). Three metrics, the anomaly correlation coefficient (ACC, see Equation 1), mean absolute error (MAE, see Equation 2), and integrated ice-edge error (IIEE, see Equation 3, Goessling et al., 2016) between observations and predictions, are used to assess the SIC prediction skill.

$$ACC = \frac{\sum_{i=1}^{n} (p_i - \overline{p}) (g_i - \overline{g})}{\sqrt{\sum_{i=1}^{n} (p_i - \overline{p})^2} \sqrt{\sum_{i=1}^{n} (g_i - \overline{g})^2}}$$
(1)

where p represents the SIC-predicted anomalies,  $p_i$  represents the prediction anomaly at time i, g represents the observed SIC anomalies. n is the length of time series.





Figure 2. Mean skill measured by (a) MAE, (b) IIEE, and (c) ACC of the different forecast models (colored lines) and the SIC anomaly persistence benchmark forecast (red-solid line) in predicting the Antarctic SIC as a function of lead time. The gray line represents the SIPNet's model skill using OSI SAF SICs as input, while all other lines represent the input of NSIDC SICs. Results have been averaged over all grid points and all target weeks from 2017 to 2021.

$$MAE = \frac{1}{n} \sum_{i=1}^{n} |P_i - G_i|$$
(2)

where  $P_i$  represents the predicted SIC value at time *i*,  $G_i$  represents the SIC observed value at time *i*.

$$IIEE = O + U \tag{3}$$

where IIEE is defined as the cumulative area encompassing discrepancies between the predicted and observed SIE, this includes both overestimated (O) and underestimated (U). We create anomaly time series from 2017 to 2021 by subtracting climatologies of the same period from weekly mean data. These metrics are only calculated in areas covered by sea ice (Figure 1c). The Antarctic is divided into five subregions based on the NSIDC classifications. In addition to the ECMWF, NCEP, and GFDL dynamical models, we also compare skill against a linear Markov model (Chen & Yuan, 2004; Wang, Yuan, Bi, et al., 2023; Wang, Yuan, Ren, et al., 2023). Contrasting to the earlier Markov models built in the multivariate space, the Markov model here only utilizes SIC to construct the model. It is worth noting that the term "target week" refers to the week being predicted, and "lead week" refers to the number of weeks prior to the target week that the forecast was initialized throughout the text.

#### 3. Results and Discussion

#### 3.1. SIPNet Versus Other Models at Subseasonal Scale

Here predictive capabilities of the SIPNet model were compared to the above-mentioned dynamical and statistical models. We also evaluated the prediction skill by using anomaly persistence forecast as a benchmark. The findings indicate substantial variation in forecast skill across various models, as depicted in Figure 2. In terms of MAE, all the dynamical models fail to exhibit any predictive skill compared to anomaly persistence, highlighting the errors in initial conditions made by the dynamical prediction systems. Zampieri et al. (2019) compared multiple S2S prediction systems for Antarctic sea ice and found that most systems tend to overestimate the SIE. The initial error in the dynamical models can have various sources, including adjustment of the sea ice edge during data assimilation, use of different sea ice observations in the assimilation and verification, and interpolation errors due to regridding (Zampieri et al., 2018).

SIPNet consistently outperforms dynamical models and the anomaly persistence model with a lower MAE across all lead times. Without modeling the interactions between the atmosphere, ice, and ocean, SIPNet considers only SIC and still achieves higher forecast skill. As sea ice variability is a product of integrated ocean-atmosphere interactions, the SIPNet can learn from the past and capture the main features of ice variability. While the Markov model outperforms most dynamic models in terms of MAE, it still has lower skill than SIPNet for all lead times. SIPNet also exhibits strong predictive capabilities for sea ice edge, with the least IIEEs consistently below  $1.78 \times 10^6$  km<sup>2</sup> across all lead times (Figure 2b). This performance is significantly superior to the dynamical models.

In terms of ACC, the dynamical models, especially ECMWF, exhibit better performance than the anomaly persistence model, which displays a rapid loss in prediction skill (Figure 2c). This suggests that the ECMWF model

captures the phase of SIC anomalies well in the Antarctic, but struggles in predicting the magnitude of SIC variability, as reflected by its large MAE. The fact that the dynamical models have some ACC skill suggests that their MAE skill has the potential to be further improved through post-forecast bias correction. GFDL-SPEAR exhibits the lowest ACC at lead times of 1–4 weeks, indicative of SIC initialization errors. Note that the GFDL-SPEAR predictions, which are developed for seasonal applications, are initialized solely on the first day of each month, resulting in only 60 prediction samples during the period from 2017 to 2021, which is much fewer than other models (at least 260 samples). This may potentially limit the forecast skill. GFDL-SPEAR demonstrates the capability of skillful predictions in regional Antarctic SIE at the seasonal timescale, surpassing a persistence forecast (Bushuk et al., 2021). SIPNet outperforms all other models, is skillful (ACC > 0.5) at 1–4 lead weeks, and showcases exceptional predictive ability in capturing both the phase and amplitude of Antarctic SIC anomalies.

We have performed additional analysis utilizing the OSI SAF SICs to assess the sensitivity of our model skill to different datasets. The results indicate that the model skill derived from both datasets is remarkably similar (Figure 2), suggesting that the differences in these two datasets are not sufficient to influence the model skill. To examine whether the model excessively relies on the length of the training set, we trained SIPNet using data from 20 years (1993–2012) and 10 years (2005–2014) separately. The results revealed that the model skill (ACC) of these two experiments only decreased by 4.4% and 7.7% compared to the model trained using 31 years (1979–2010) of data (Figure S1 in Supporting Information S1). This suggests that the length of the training set only slightly influences the model skill. We have also investigated the model's predictive capacity for the extreme of SIE in February and September. Figures S2a and S2b in Supporting Information S1 reveals that the model's ability to forecast extreme SIE decreases as lead time increases. However, the forecast error remains relatively small at the 1–3 lead weeks, especially during the rapid decline events in 2015/2016 when the predicted SIE error is  $0.082 \times 10^6$  km<sup>2</sup> at a 3-week lead in February. The predicted error of extreme SIE also does not depend on the SIE anomaly (Figures S2c and S2d in Supporting Information S1). Next, we examine the spatial and seasonal distributions of SIPNet's prediction skill.

#### 3.2. Seasonal and Regional Features of Model Skill

Since ECMWF performs better than other dynamic models in Antarctic sea ice predictions, here we use it to represent dynamic models. We present the 4-week leads predictions from SIPNet, Markov, and ECMWF models to reveal prediction skills' seasonal and spatial features (Figure 3 and Figure S3 in Supporting Information S1). Consistent with earlier work (Bushuk et al., 2021), our results indicate that the predictability of Antarctic sea ice is seasonally dependent, according to different physical conditions. During autumn, the SIPNet exhibits its highest level of performance, evidenced by its highest ACC (Figure 3a) and lowest MAE (Figure S3 in Supporting Information S1). This superior performance could be related to the physical mechanism of fall predictability provided by predictable ocean heat content anomalies (Holland et al., 2013), presumably reflected in the initial SIC field. Conversely, the SIPNet's lowest skill is observed during spring, where it registers the lowest ACC and highest MAE. The forecasting error during spring mainly occurs in the sea ice-marginal zone. However, the model's ACC is still significant as it can skillfully capture phases of sea ice anomalies despite having a relatively large forecasting error for the magnitudes of those anomalies. During the cold season, SIC encircling the Antarctic continent away from the ice edge is nearly 100%, with limited variability, resulting in relatively low ACC. However, the model is adept at capturing sea ice climatology in these areas, so the MAE is small with less than 6%. The MAE is less than 18%, even in the West Pacific, significantly less than 20%-35% of SIC standard deviation (Figure S4 in Supporting Information S1).

SIPNet exhibits similar skill patterns as the Markov model across all seasons, especially during the cold season (Figures 3a and 3b). Compared to the Markov model, SIPNet has higher skill in autumn Amundsen and Weddell Seas, where the Antarctic Dipole (ADP) is located (Figure 3d). It also reflects the limitations of the Markov model in predicting the principal components under fixing spatial patterns of the coupled climate modes, which do not represent sea ice's synchronized spatiotemporal evolution. Additionally, ADP is the dominant variability of Antarctic sea ice, driven by ENSO and the Antarctic air-sea interactions. The high skill in these regions suggests that SIPNet may be able to capture the remote climate-forcing signal on sea ice, particularly in fall (Figure S3d in Supporting Information S1). In addition, Figure 3 and Figure S3 in Supporting Information S1 illustrate a notable disparity in the skill patterns of SIPNet and ECMWF. Compared to ECMWF, SIPNet primarily demonstrates reduced MAE across all seasons, especially in spring, as depicted by green shades in Figure S3e in Supporting Information S1.



# **Geophysical Research Letters**



Figure 3. Model skills measured by ACC between SIC predictions at 4-week leads and observations from 2017 to 2021 as a function of target seasons (a–c). ACC difference between SIPNet and Markov (d), as well as between SIPNet and ECMWF (e).

To further quantify the model skill in each region, we conducted spatial averaging of MAE and ACC at lead times of 1–8 weeks in Figure 4a and Figure S5 in Supporting Information S1, respectively. The results show that SIPNet outperforms persistence in terms of MAE and ACC across all regions at all lead times. The Weddell Sea shows the best performance, with the smallest MAE (less than 8%) even at a lead time of 8 weeks in the cold season and also maintaining a low level of MAE in the warm season. Its ACC also leads other regions, except for the



# **Geophysical Research Letters**



**Figure 4.** (a) The seasonal and regional averaged model skills are measured by the MAE between prediction and observation from 2017 to 2021 as a function of lead months for each Antarctic region. The average MAE for 1–8 week leads from SIPNet is marked in the lower right corner of each panel. (b) Linear regression coefficients of ENSO and SAM on SIC predictions at 4-week leads. Panel (c) is the same as (b) except for SIC observations. Patterns in (b) and (c) represent anomalies associated with one standard deviation of the corresponding index. The blue (red) circle marked with L (H) in (b) represents the center of negative (positive) pressure anomaly as an El Niño -like PSA pattern. The red (blue) arrow represents the warm (cold) advection driven by PSA. The PSA pattern consists of a positive anomalous sea level pressure (SLP) center in the Amundsen Sea and negative SLP anomalies to the east of New Zealand and in South America/South Atlantic when El Niño dominated the tropical Pacific.

Amundsen and Bellingshausen Sea (Figure S5 in Supporting Information S1). The Markov and ECMWF also outperform the persistence in the Weddell Sea during summer and winter. Several factors could contribute to the high predictability of sea ice in the Weddell Sea. First, the high variability of sea ice associated with ENSO and the Amundsen Sea Low (ASL) could provide crucial SIC predictability. Additionally, the Weddell Sea dominates the multiyear ice coverage of the Southern Ocean and has a longer SIC persistence due to its thick ice. The steady eastward sea ice drift may be another contributing factor (Bushuk et al., 2021).

The Indian Ocean displays a smaller MAE (below 5%) even at an 8-week lead time (Figure 4a) and a higher ACC (Figure S5 in Supporting Information S1) in autumn. The MAE in the Ross Sea is lower in winter but higher in summer. The high variability of Ross sea ice does not contribute much to the overall model's skill, possibly due to the complex physical processes involved. The West Pacific exhibits larger MAE and lower ACC across all seasons, likely because the region lacks strong atmospheric circulation that drives sea ice, leading to a lower signal-to-noise ratio in sea ice variability. Additionally, the West Pacific is characterized predominantly by thin ice and seasonal ice, leading to shorter ice persistence. In summary, our SIPNet model performs best in the Weddell Sea and imperfectly in the West Pacific.

In addition, SIPNet exhibits higher ACC than ECMWF in most regions, except for the spring Weddell Sea and West Pacific (Figure S5 in Supporting Information S1). It also demonstrates better ACC than Markov in most regions, except for the autumn Ross Sea at the 5–8 week leads. All three models concur that the Amundsen/Bell-ingshausen Sea exhibits the most significant forecast errors. Interestingly, they all also demonstrate the highest ACC in this region (Figure 4a and Figure S5 in Supporting Information S1), indicating that these models perform well in capturing SIC anomaly signals, but struggle slightly with the magnitudes of anomalies in the region. The intricate physical mechanisms underlying this phenomenon merit further investigation, yet are beyond the scope of the present study. Moreover, whether the high ACC in the Amundsen and Weddell Seas reflects SIPNet's ability to capture the signal of climate forcing on sea ice requires further discussion.

We conducted a regression analysis of the ENSO and SAM index on the prediction and observation anomalies from 2017 to 2021 (Figures 4b and 4c). El Niño events generate stationary Rossby wave trains that curve poleward (Ding et al., 2012; Meehl et al., 2016; Yuan, 2004) and is known as the PSA pattern (Mo & Higgins, 1998), weakening the ASL (Simpkins et al., 2016). The offshore wind associated with the ASL contributes to the formation of more sea ice over the Weddell Sea through both cold atmospheric advection and offshore ice drift (Li et al., 2014; Meehl et al., 2019; Raphael et al., 2016). Conversely, onshore winds melt and compress sea ice around the Amundsen and Bellingshausen Sea region, thus generating out-of-phase sea ice anomalies similar to the ADP pattern (Yuan, 2004). Although SAM is the leading mode of extratropical Southern Hemisphere atmospheric circulation (Thompson & Wallace, 2000), its influence on sea ice is relatively weaker than ENSO-related PSA (Yuan & Li, 2008). During a positive SAM phase, westerly winds intensify, producing a sea ice anomaly pattern opposite to that related to El Niño (Figure 4b), consequently weakening sea ice variability and reducing the strength of the ADP. The anomaly patterns presented in Figures 4b and 4c are highly similar, demonstrating that the SIPNet model can well capture sea ice's responses to these forcings.

### 4. Concluding Remarks

In this study, we developed a DL model (called SIPNet) to forecast Antarctic SIC at the subseasonal timescale. To assess the model's predictive capability, we compared its performance against advanced dynamical models such as ECMWF, NCEP, and GFDL-SPEAR, as well as a linear Markov model.

We found that all the dynamical models fail to exhibit any predictive skill compared to anomaly persistence in terms of MAE. The Markov model has limits to predicting the synchronized spatiotemporal evolution of sea ice because it retains fixed spatial patterns of the coupled climate modes. By efficiently capturing the spatiotemporal features of sea ice and bypassing the intricate physical mechanisms, SIPNet outperforms persistence in terms of both MAE and ACC across all seasons and regions, at all lead times. Seasonally, SIPNet performed best in autumn with the lowest MAE and the highest ACC, and worst in spring. Spatially, the Weddell Sea performed best, with the smallest MAE (less than 9%) even at an 8-week lead time and a significant ACC. While the West Pacific is the worst-predicted region with a larger MAE and lower ACC. In addition, despite the opposite SIC anomaly patterns related to ENSO and SAM from 2017 to 2021, which result in a weaker ADP pattern, SIPNet can still capture these forcing signals well. Furthermore, we demonstrated that the model skill of SIPNet is not highly sensitive to the length of the training set time series. SIPNet exhibits well capabilities in predicting sea ice edges and extreme sea ice events.

Overall, SIPNet, showcasing exceptional predictive ability in capturing both the phase and amplitude of Antarctic SIC anomalies, outperforms ECMWF, NCEP, GFDL-SPEAR, and Markov models at a subseasonal timescale. By employing an encoder-decoder structure considering residual CNN (ResNet) with TSAM, SIPNet can capture the intricate nature of sea ice variability, including nonlinear characteristics and spatiotemporal dependencies,

the potential to further improve the model's performance.

19448007, 2023, 17, Downloaded from https rary.wiley.com/doi/10.1029/2023GL104347 by Cochrane Germany, Wiley Online Library on [31/08/2023] and on Wiley Online Library OA articles

Acknowledgments This work is supported by the Natural Science Foundation of Shandong Province, China (ZR2021QD059); National Natural Science Foundation of China (42106223 and 42206202); China Postdoctoral Science Foundation (2020TQ0322); and Strategic Priority Research Program of the Chinese Academy of Sciences (XDB4200000). Yuan is supported by the Lamont-Doherty Earth Observatory of Columbia University and a private donation (Ying).

### Data Availability Statement

The result data of this paper are available at the figshare website (https://doi.org/10.6084/m9.figshare.23691429, Wang, Yuan, Ren, et al., 2023). Daily SICs from NSIDC are available at https://doi.org/10.5067/7Q8HCCWS4I0R. Daily SICs from OSI SAF are available at https://osi-saf.eumetsat.int/products/sea-ice-products. The SAM index is available at http://www.nerc-bas.ac.uk/icd/gjma/sam.html. The ENSO index is available at https://psl.noaa.gov/gcos\_wgsp/Timeseries/Nino34/. The SIC predictions from ECMWF and NCEP are obtained from the database of the S2S Prediction project at https://apps.ecmwf.int/datasets/data/s2s-realtime-instantaneous-accum-ecmf/levtype=sfc/type=cf/.

while bypassing complex model physics and parameterization. SIPNet also has advantages such as requiring minimal computing resources and fast computation speed. The model effectively fills the gap in Antarctic SIC prediction at the subseasonal timescale. In our future investigation, we will explore the possibility of enhancing the model's skill by incorporating various climate variables, including SIT, ocean heat content (OHC), sea surface temperature (SST), air temperature (AT), wind, and atmospheric pressure fields. Additionally, we will also consider the influence of climate factors such as ENSO, ASL, SAM, and wave-3. These enhancements hold

#### References

- Andersson, T. R., Hosking, J. S., Perez-Ortiz, M., Paige, B., Elliott, A., Russell, C., et al. (2021). Seasonal Arctic sea ice forecasting with probabilistic deep learning. *Nature Communications*, 12(1), 5124. https://doi.org/10.1038/s41467-021-25257-4
- Bushuk, M., Winton, M., Haumann, F. A., Delworth, T., Lu, F., Zhang, Y., et al. (2021). Seasonal prediction and predictability of regional Antarctic sea ice. *Journal of Climate*, 34(15), 6207–6233. https://doi.org/10.1175/jcli-d-20-0965.1
- Chen, D., & Yuan, X. (2004). A Markov model for seasonal forecast of Antarctic sea ice. *Journal of Climate*, 17(16), 3156–3168. https://doi.org/ 10.1175/1520-0442(2004)017<3156:AMMFSF>2.0.CO;2
- Chi, J., & Kim, H.-C. (2017). Prediction of arctic sea ice concentration using a fully data driven deep neural network. *Remote Sensing*, 9(12), 1305. https://doi.org/10.3390/rs9121305
- Comiso, J. C. (2017). Bootstrap sea ice concentrations from nimbus-7 SMMR and DMSP SSM/I-SSMIS, version 3. NASA National Snow and Ice Data Center Distributed Active Archive Center. https://doi.org/10.5067/7Q8HCCWS4I0R
- Comiso, J. C., Gersten, R. A., Stock, L. V., Turner, J., Perez, G. J., & Cho, K. (2017). Positive trend in the Antarctic sea ice cover and associated changes in surface temperature. *Journal of Climate*, 30(6), 2251–2267. https://doi.org/10.1175/jcli-d-16-0408.1
- Delworth, T., Cooke, W., Adcroft, A., Bushuk, M., Chen, J.-H., Dunne, K., et al. (2020). SPEAR: The next generation GFDL modeling system for seasonal to multidecadal prediction and projection. *Journal of Advances in Modeling Earth Systems*, *12*(3), e2019MS001895. https://doi. org/10.1029/2019MS001895
- Ding, Q., Steig, E. J., Battisti, D. S., & Wallace, J. M. (2012). Influence of the tropics on the southern annular mode. *Journal of Climate*, 25(18), 6330–6348. https://doi.org/10.1175/jcli-d-11-00523.1
- Fogt, R. L., Sleinkofer, A. M., Raphael, M. N., & Handcock, M. S. (2022). A regime shift in seasonal total Antarctic sea ice extent in the twentieth century. *Nature Climate Change*, 12(1), 54–62. https://doi.org/10.1038/s41558-021-01254-9
- Goessling, H. F., Tietsche, S., Day, J. J., Hawkins, E., & Jung, T. (2016). Predictability of the Arctic sea ice edge. *Geophysical Research Letters*, 43(4), 1642–1650. https://doi.org/10.1002/2015g1067232
- Handcock, M. S., & Raphael, M. N. (2020). Modeling the annual cycle of daily Antarctic sea ice extent. *The Cryosphere*, 14(7), 2159–2172. https://doi.org/10.5194/tc-14-2159-2020
- Holland, M. M., Blanchard-Wrigglesworth, E., Kay, J., & Vavrus, S. (2013). Initial-value predictability of Antarctic sea ice in the community climate system model 3. *Geophysical Research Letters*, 40(10), 2121–2124. https://doi.org/10.1002/grl.50410
- Kim, Y. J., Kim, H. C., Han, D., Lee, S., & Im, J. (2020). Prediction of monthly Arctic sea ice concentrations using satellite and reanalysis data based on convolutional neural networks. *The Cryosphere*, 14(3), 1083–1104. https://doi.org/10.5194/tc-14-1083-2020
- Leppäranta, M., Meleshko, V. P., Uotila, P., & Pavlova, T. (2020). *Sea ice in the Arctic*. Springer. https://doi.org/10.1007/978-3-030-21301-5\_8 Li, X., Cai, W., Meehl, G. A., Chen, D., Yuan, X., Raphael, M., et al. (2021). Topical teleconnection impacts on Antarctic climate changes.
- Nature Reviews Earth & Environment, 2(10), 680–698. https://doi.org/10.1038/s43017-021-00204-5
  Li, X., Holland, D. M., Gerber, E. P., & Yoo, C. (2014). Impacts of the north and tropical Atlantic Ocean on the Antarctic Peninsula and sea ice.
  Nature, 505(7484), 538–542. https://doi.org/10.1038/nature12945
- Liu, Y., Bogaardt, L., Attema, J., & Hazeleger, W. (2021). Extended-range arctic sea ice forecast with convolutional long short-term memory networks. *Monthly Weather Review*, 149(6), 1673–1693. https://doi.org/10.1175/mwr-d-20-0113.1

Marchi, S., Fichefet, T., Goosse, H., Zunz, V., Tietsche, S., Day, J. J., & Hawkins, E. (2019). Reemergence of Antarctic sea ice predictability and its link to deep ocean mixing in global climate models. *Climate Dynamics*, 52(5–6), 2775–2797. https://doi.org/10.1007/s00382-018-4292-2

Marshall, G. J. (2003). Trends in the southern annular mode from observations and reanalyses. *Journal of Climate*, *16*(24), 4134–4143. https://doi.org/10.1175/1520-0442(2003)016<4134:TITSAM>2.0.CO;2

- Massonnet, F., Reid, P., Lieser, J. L., Bitz, C. M., Fyfe, J., & Hobbs, W. R. (2022). Assessment of summer 2021-2022 sea-ice forecasts for the Southern Ocean *Technical Note*. Retrieved from https://fmassonn.github.io/sipnsouth.github.io/
- Meehl, G. A., Arblaster, J. M., Bitz, C. M., Chung, C. T. Y., & Teng, H. (2016). Antarctic sea-ice expansion between 2000 and 2014 driven by tropical Pacific decadal climate variability. *Nature Geoscience*, 9(8), 590–595. https://doi.org/10.1038/ngeo2751
- Meehl, G. A., Arblaster, J. M., Chung, C. T., Holland, M. M., DuVivier, A., Thompson, L., et al. (2019). Sustained ocean changes contributed to sudden Antarctic sea ice retreat in late 2016. *Nature Communications*, 10(1), 14. https://doi.org/10.1038/s41467-018-07865-9

Mo, K. C., & Higgins, R. W. (1998). The Pacific-South American modes and tropical convection during the Southern Hemisphere winter. Monthly Weather Review, 126(6), 1581–1596. https://doi.org/10.1175/1520-0493(1998)126<1581:tpsama>2.0.co;2 by the

- Morioka, Y., Iovino, D., Cipollone, A., Masina, S., & Behera, S. (2021). Summertime sea-ice prediction in the Weddell Sea improved by sea-ice thickness initialization. *Scientific Reports*, 11(1), 11475. https://doi.org/10.1038/s41598-021-91042-4
- Parkinson, C. L., & Cavalieri, D. J. (2012). Antarctic sea ice variability and trends, 1979-2010. The Cryosphere, 6(113), 871–880. https://doi. org/10.5194/tc-6-871-2012
- Parkinson, C. L. (2019). A 40-y record reveals gradual Antarctic sea ice increases followed by decreases at rates far exceeding the rates seen in the Arctic. Proceedings of the National Academy of Sciences of the United States of America, 116(29), 14414–14423. https://doi.org/10.1073/ pnas.1906556116
- Raphael, M. N., Marshall, G. J., Turner, J., Fogt, R. L., Schneider, D., Dixon, D. A., et al. (2016). The Amundsen Sea low: Variability, change, and impact on Antarctic climate. Bulletin of the American Meteorological Society, 97(1), 197–210. https://doi.org/10.1175/BAMS-D-14-00018.1
- Rayner, N. A., Parker, D. E., Horton, E. B., Folland, C. K., Alexander, L. V., Rowell, D. P., et al. (2003). Global analyses of sea surface temperature, sea ice, and night marine air temperature since the late nineteenth century. *Journal of Geophysical Research*, 108(D14), 4407. https:// doi.org/10.1029/2002JD002670
- Ren, Y., & Li, X. (2023). Predicting the daily sea ice concentration on a subseasonal scale of the pan-arctic during the melting season by a deep learning model. *IEEE Transactions on Geoscience and Remote Sensing*, 61, 1–15. https://doi.org/10.1109/tgrs.2023.3279089
- Ren, Y., Li, X., & Zhang, W. (2022). A data-driven deep learning model for weekly sea ice concentration prediction of the Pan-Arctic during the melting season. *IEEE Transactions on Geoscience and Remote Sensing*, *60*, 1–19. https://doi.org/10.1109/TGRS.2022.3177600
- Roach, L. A., Doerr, J., Holmes, C. R., Massonnet, F., Blockley, E. W., Notz, D., et al. (2020). Antarctic sea ice area in CMIP6. *Geophysical Research Letters*, 47(9), e2019GL086729. https://doi.org/10.1029/2019gl086729
- Simpkins, G. R., Peings, Y., & Magnusdottir, G. (2016). Pacific influences on tropical Atlantic teleconnections to the southern Hemisphere high latitudes. Journal of Climate, 29(18), 6425–6444. https://doi.org/10.1175/jcli-d-15-0645.1
- Sørensen, A. M., Lavergne, T., Eastwood, S., & Norway, M. (2022). Global sea ice concentration climate data records product user's manual. https://doi.org/10.15770/EUM\_SAF\_OSI\_0014
- Thompson, D. W., & Wallace, J. M. (2000). Annular modes in the extratropical circulation. Part I: Month-to-month variability. Journal of Climate, 13(5), 1000–1016. https://doi.org/10.1175/1520-0442(2000)013<1000:AMITEC>2.0.CO;2
- Vitart, F., Ardilouze, C., Bonet, A., Brookshaw, A., Chen, M., Codorean, C., et al. (2017). The subseasonal to seasonal (S2S) prediction project database. *Bulletin of the American Meteorological Society*, 98(1), 163–173. https://doi.org/10.1175/bams-d-16-0017.1
- Wang, Y., Yuan, X., Bi, H., Bushuk, M., Liang, Y., Li, C., & Huang, H. (2022). Reassessing seasonal sea ice predictability of the Pacific-Arctic sector using a Markov model. *The Cryosphere*, 16(3), 1141–1156. https://doi.org/10.5194/tc-16-1141-2022
- Wang, Y., Yuan, X., Bi, H., Ren, Y., Liang, Y., Li, C., & Li, X. (2023). Understanding Arctic sea ice thickness predictability by a Markov model. Journal of Climate, 36(15), 4879–4897. https://doi.org/10.1175/JCLI-D-22-0525.1
- Wang, Y., Yuan, X., Ren, Y., Bushuk, M., Shu, Q., Li, C., & Li, X. (2023). Subseasonal prediction of regional Antarctic sea ice by a deep learning model: Supplemental data. figshare. Journal contribution. https://doi.org/10.6084/m9.figshare.23691429.v2
- Yuan, X. (2004). ENSO-Related impacts on Antarctic sea ice: A synthesis of phenomenon and mechanisms. Antarctic Science, 16(4), 415–425. https://doi.org/10.1017/S0954102004002238
- Yuan, X., Kaplan, M. R., & Cane, M. A. (2018). The interconnected global climate system—A review of tropical–polar teleconnections. Journal of Climate, 31(15), 5765–5792. https://doi.org/10.1175/JCLI-D-16-0637.1
- Yuan, X., & Li, C. (2008). Climate modes in southern high latitudes and their impacts on Antarctic sea ice. Journal of Geophysical Research, 113(C6), C06S91. https://doi.org/10.1029/2006JC004067
- Zampieri, L., Goessling, H. F., & Jung, T. (2018). Bright prospects for arctic sea ice prediction on subseasonal time scales. Geophysical Research Letters, 45(18), 9731–9738. https://doi.org/10.1029/2018gl079394
- Zampieri, L., Goessling, H. F., & Jung, T. (2019). Predictability of Antarctic sea ice edge on subseasonal time scales. Geophysical Research Letters, 46(16), 9719–9727. https://doi.org/10.1029/2019GL084096
- Zhang, L., Delworth, T. L., Yang, X., Zeng, F., Lu, F., Morioka, Y., & Bushuk, M. (2022). The relative role of the subsurface Southern Ocean in driving negative Antarctic Sea ice extent anomalies in 2016-2021. *Communications Earth & Environment*, 3(1), 302. https://doi.org/10.1038/ s43247-022-00624-1